Author's response to reviews

Title: A case presentation of postoperative neurological aggravation in a patient with Xeroderma Pigmentosum: Sevoflurane is it responsible?

Authors:

salaheddine fjouji (salaheddinefj@hotmail.com)
musatpha bensghir (mustaphabens_15rea@hotmail.com)
bahija yafat (yafat.b@yahoo.fr)
najib bouhabba (n.bouhabba@hotmail.com)
lhoucine boutayeb (boutayebl@yahoo.fr)
hicham azendour (hichamazendour@hotmail.com)
nordine drissi kamili (nordinedriss@hotmail.com)

Version: 2 Date: 3 June 2012

Author's response to reviews: see over
Dear editorial board of Journal of Medical Case reports:

Please find enclosed the manuscript entitled: A case presentation of postoperative neurological aggravation in a patient with Xeroderma Pigmentosum: Sevoflurane is it responsible? A case presentation by Mustapha Bensghir and al to be submitted as a case presentation to Journal of Medical Case reports for consideration of publication. All co-authors have seen agree with the contents of the manuscript and there is no financial interest to report. We certify that the submission is original work and is not under review at any other publication. In this manuscript, we report an unusual case presentation of post operative neurological aggravation in a patient with Xeroderma Pigmentosum anesthetized by sevoflurane. We believe that this case presentation could be interest to the readers of Journal of Medical Case reports because we discussed the mechanism of toxicity of inhalation anesthetic agents for patient with Xeroderma Pigmentosum and the relation between this agent and this disease evolution.

We report this case presentation to draw attention to clinical practitioners about the risk of neurological deterioration of patients with XP by anesthetic gases, and we suggest using intravenous anesthetics in such patients. We hope that the editorial board will agree on the interest of this study

Sincerely yours Mustapha

Bensghir M.D. on behalf of the authors